Author's response to reviews

Title: Antithrombotic treatment in elderly patients with atrial fibrillation: a practical approach

Authors:

   Carmen Suarez Fernandez (csuarezf@salud.madrid.org)
   Francesc Formiga (fformiga@bellvitgehospital.cat)
   Miguel Camafort (miguel.camafort@gmail.com)
   Jose María Cepeda Rodrigo (jmcepedarodrigo@gmail.com)
   Jesús Díez-Manglano (jdiez@aragon.es)
   Antonio Pose Reino (antonioposereino@gmail.com)
   Gregorio Tiberio (gregorio.tiberio@unavarra.es)
   Jose María Mostaza (josemaria.mostaza@salud.madrid.org)

Version: 2 Date: 30 April 2015

Author's response to reviews:

Dear editor,

As requested the article has been revised:
- Line and page numbers have been added.
- The abstract has been structured.
- Sections have been amended as requested.
- Competing Interests, authors' Contributions and acknowledgements have been specified.

Thank you very much for your attention.